Week 22 starting 29 May 06
SM18 Operation Meeting (Tuesdays) 30 May 06

Bench status  on Monday  29-05-2006 at: 5/29/2006 10:22:26 AM    
	7
	A1
	MBBL1236
	Min Energy @ 6500
	5.47

	9
	A2
	MBAR1354
	PTE Min. NRJ 3KA Diode Test
	1.90

	6
	B1
	SSS264
	PT 13.2 Resist. Meas.
	298.94

	12
	B2
	SSS505
	PT 14 Magnet MAPS
	300.82

	10
	C1
	MBAR2310
	Prep 5 Cool Down
	98.46

	8
	C2
	MBAR2272
	PT 6.4 Training 4
	1.90

	4
	D1
	SSS647
	MM - Stretched wire at 4.4K
	2.35

	1
	D2
	SSS677
	PT 14 Magnet MAPS
	299.83

	11
	E1
	MBAR2276
	ICS 4 Final connection
	300.37

	5
	E2
	MBAR1353
	PT 9 CDAP HV
	2.35

	3
	F1
	MBBR2279
	PT 14 Magnet MAPS
	299.83

	2
	F2
	MBAR2327
	ICS 3 Traveller after warm test
	299.92


  CONFIG : 
 D8- AS2 –IRSSS2;
Supervisor: Pierre/        Eq support : 16-3602 :

14 Magnet Tests completed :  10 Dipoles (),    2 Arc-SSS  ( no   REPEAT)

(14 Magnets in all)

0  500-series ( 524 :)  

(2255 was a  mistake)             2  600-series [ 628 (134 hrs),  663 (114 hrs) ]

General Points :

· Last weekend problems  ( (Eric) : F2 2279/

· A & F clusters both cold due to problems on 2279 & 1236 (MM)

· D1 problems powering ( IAP adjust) SSW, Shafts delay etc

· D2 Special M’ments & switching back & forth?

· 505 could not be removed since last Friday ? (Rocla, Transport etc)

· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· News /Situation from (SSS, SSSS)  :  J Halik
· Situation SSSS  :OP issues : 
· A.O.B.

SITUATION ON SATURDAY EVENING   at 20.45 hrs  27th May 2006; 

 TOO Many Things all cold same time;   A1, A2;   F1, F2;   D1 ,D2
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Tests 

	Magnets under Test from 21-05-2006 to 27-05-2006 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBR2304 
	[image: image2.png]



	


	19-05-2006 
	24 h.
	20 h.
	24 h.
	15 h.
	16 h.
	23-05-2006 
	99 h.
	Stripping
	
	

	2
	TBA2
	MBBR2277 
	[image: image4.png]



	


	19-05-2006 
	21 h.
	28 h.
	37 h.
	22 h.
	10 h.
	24-05-2006 
	118 h.
	Stripping
	
	

	3
	TBB1
	SSS263 
	[image: image6.png]



	


	21-05-2006 
	31 h.
	15 h.
	23 h.
	9 h.
	12 h.
	25-05-2006 
	90 h.
	Stripping
	
	

	4
	TBB1
	SSS265 
	[image: image8.png]



	


	17-05-2006 
	29 h.
	19 h.
	18 h.
	10 h.
	32 h.
	21-05-2006 
	108 h.
	Stripping
	
	

	5
	TBB2
	SSS628 
	[image: image10.png]



	


	16-05-2006 
	31 h.
	18 h.
	51 h.
	11 h.
	23 h.
	22-05-2006 
	134 h.
	Stand By
	
	

	6
	TBC1
	MBAL1393 
	[image: image12.png]



	


	18-05-2006 
	29 h.
	20 h.
	23 h.
	18 h.
	27 h.
	23-05-2006 
	117 h.
	Stripping
	
	

	7
	TBC1
	MBBR2315 
	[image: image14.png]



	


	23-05-2006 
	32 h.
	23 h.
	19 h.
	14 h.
	7 h.
	27-05-2006 
	95 h.
	Stripping
	
	

	8
	TBC2
	MBBR2287 
	[image: image16.png]



	


	17-05-2006 
	24 h.
	24 h.
	21 h.
	18 h.
	8 h.
	21-05-2006 
	95 h.
	Stripping
	
	

	9
	TBC2
	MBBR2291 
	[image: image18.png]



	


	21-05-2006 
	29 h.
	20 h.
	17 h.
	15 h.
	23 h.
	26-05-2006 
	104 h.
	Stripping
	
	

	10
	TBD2
	SSS663 
	[image: image20.png]



	


	17-05-2006 
	33 h.
	18 h.
	25 h.
	10 h.
	28 h.
	22-05-2006 
	114 h.
	Stand By
	
	

	11
	TBE1
	MBAL1388 
	[image: image22.png]



	


	18-05-2006 
	24 h.
	20 h.
	19 h.
	16 h.
	20 h.
	22-05-2006 
	99 h.
	Stripping
	
	

	12
	TBE1
	MBBR2255 
	[image: image24.png]



	


	22-05-2006 
	9 h.
	0 h.
	0 h.
	0 h.
	16 h.
	23-05-2006 
	25 h.
	Stand By
	
	

	13
	TBE2
	MBAR2275 
	[image: image26.png]



	


	20-05-2006 
	24 h.
	21 h.
	16 h.
	15 h.
	34 h.
	25-05-2006 
	110 h.
	Stripping
	
	

	14
	TBF1
	MBBR2321 
	[image: image28.png]



	


	19-05-2006 
	22 h.
	22 h.
	14 h.
	15 h.
	9 h.
	22-05-2006 
	82 h.
	Stripping
	
	

	15
	TBF2
	MBBR2305 
	[image: image30.png]



	


	20-05-2006 
	28 h.
	22 h.
	16 h.
	15 h.
	16 h.
	24-05-2006 
	97 h.
	Stripping
	
	


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBR2304
	10294.9 A.
(D1_L)
	11575.3 A.
(D2_U)
	12026.8 A.
(D2_L)
	12427.8 A.
(D1_U)
	12850 A.
[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBR2277
	10223.2 A.
(D1_U)
	11814.4 A.
(D2_U)
	12296.8 A.
(D2_L)
	12442 A.
(D1_U)
	[image: image33.png]



	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS263
	11833.7 A.
(QE_12)
	11982.5 A.
(QE-34)
	12282.3 A.
(QE_12)
	12000 A.
(No Data)
	12000 A.
(No Data)
	12504 A.
(QE-34)
	[image: image34.png]



	-
	-
	-
	-
	-
	-

	4
	SSS265
	11121.8 A.
(QI-34)
	12082.7 A.
(QI_12)
	12234 A.
(QE_12)
	[image: image35.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS628
	5692.1 A.
(E_MQML12)
	5820 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAL1393
	10553.3 A.
(D1_L)
	11615 A.
(D2_U)
	12315.7 A.
(D2-U)
	12533.1 A.
(D1-U)
	[image: image37.png]



	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBBR2315
	12374.3 A.
(D1_L)
	12256.6 A.
(D2-U)
	12800.5 A.
(D1_L)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBBR2287
	9830 A.
(D1_L)
	11859.3 A.
(D1_U)
	12268.5 A.
(D1_U)
	12850 A.
[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBR2291
	11198.6 A.
(D1_L)
	11851.3 A.
(D2_L)
	12772.1 A.
(D1_L)
	12850 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS663
	5582 A.
(MQMC_EXT)
	5850 A.
[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL1388
	11529.2 A.
(D1_L)
	12204 A.
(D1-L)
	12544 A.
(D1-U)
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBR2255
	[image: image43.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAR2275
	10698.9 A.
(D!-U)
	12221 A.
(D1_U)
	12714 A.
(D2_L)
	[image: image44.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBBR2321
	12491 A.
(D1_L)
	12850 A.
[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBR2305
	11112.7 A.
(D2_L)
	12219.1 A.
(D2_U)
	12557.4 A.
(D1_U)
	[image: image46.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-


 

[image: image47.png]Magnets from Sunday, May 21, 2006 to Saturday, May 27, 2006
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                          2255 brought by mistake ;     so only 14 magnets in whole week due to problems & holiday w’end (ascension) :








( SSSS & powering +SSW,  special tests .. etc)
[image: image48]
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_23_M
	TBB1
	SSS263 : PT2 HV-CQBP fails (Quadrupole vs Ground shows high current > 10A).
	MTM
	
	Richard advised to re-launch the test and check.

	Tuesday_23_M
	TBD2
	SSS677: Lyre test problem: current 2A at 15V not possible.
	MTM
	
	Richard advised to conduct test at 15V,1A. Final test was OK

Action : Define a global rule for S4: i.e. 1A and 15 V

	Tuesday_23_M
	TBB1
	SSS263: Magnet name not appearing in TEMA. So proper test sequence could not be selected.
	MTM
	TEMA
	Richard advised to re-launch TEMA application. The problem was solved.

	Tuesday_23_A
	TBB1
	SSS263 : Communication error with PLC & LF.
	AB/CO
	
	Closing all lab-view applications and re-launching them was not sufficient to get it restarted. 

Communication error was solved by shutting down the Sun machine. 

	Tuesday_23_N
	TBF2 
	MB2305 : IAP@warm. LF DAQ did not trigger. 
	AB/CO
	lf
	OP team reset all the crates and Suns, TEMA application is stuck with "Error receiving info from udp socket". IAP test was executed locally.

	Tuesday_23_N
	TBB1
	SSS263: MQ quenched twice on the 12kA plateau during PT8 test. Cryo will increase the flow in the bus-bars for the next attempt. 
	ACR
	
	

	Wednesday_24_M

Thursday_25_A

Saturaday_27_M
	TBB1

TBB2

TBB1
	HV panel is not opening at control room.

Same problem during HV test on SSS505.

Same problem for HV test on SSS264
	IT
	HV
	Network traffic increased?

HV panel aperture is set in low priority.

	Wednesday_24_M
	TBF
	Communication problem coming from UDP socket. 
	IT?
	
	Richard contacted and had to reboot TEMA in Sun machine. Traffic in the network?

	Wednesday_24_M
	TBB2
	SSS505 : PT4 could not be done because of cryo instability
	ACR
	
	Why?

	Wednesday_24_A
	TBD1
	SSS647: Error in HV while doing step related to orbit dipole. 

Errors observed in Temperature sensor and Cryo heater steps also in the report but not while performing tests.
	MTM
	
	Richard asked to skip step.

 Richard will correct the report and generate NC as required.

	Wednesday_24_A
	TBB2
	SSS505: Corrector powering EF could not be launched due to trigger condition. Origin : NO CD circuit in 5XX series.
	MTM
	
	Richard cleared the triggers and restarted the system. Corrector powering could be carried out subsequently.

	Wednesday_24_A
	TBD1
	SSS647: Discrepancy between the values read by TT821-PT147 sensors.
	ACR


	
	Richard asked to ignore PT147  as this sensor is not calibrated down to 4.5K Action : TT821 will be compared with PT148.

	Wednesday_24_A
	TBB2
	SSS505: Corrector EF powering: 10 quenches done but still the corrector is not trained. Detraining observed in 9th and 10 quenches (from 590 to 540A) Files not generated for the 9th and 10th quench.
	MTM
	
	TC informed. He asked to perform one more training. 

Test was repeated multiple times in the afternoon shift. Marco was contacted and confirmed that test can be considered successful as current is greater than 400 A. 

	Thursday_25_M
	TBB2
	SSS505: Auxiliary Current lead temperature problem, OP team is not able to launch PT5.8 corrector powering (MN). Cryo has been informed.
	ACR
	
	Solved by ACR.

	Thursday_25_A
	TBD1
	SSS647: SSW could not be initiated because of  an interlock. Power converter could not be put on "ON STANDBY". 
	MTM
	
	Richard contacted. He found out that the interlock came from the offsets of some cards for the other magnet in the cluster (MQM). Origin : Wrong sequence  selected.

With the correct sequence the problem persists. Richard had also redone the AC compensation, on the first 2 Potaim cards, in order to be able to switch ON the power converter. Origin non symmetrical inductance in one magnet of the test bench. Action: For S4, AC compensations have to be r-done when there is a switch for one bench to another.

	Thursday_25_N
	TBB1
	SSS264 is shown on the e-traveller in place of SSS 269. Magnet entered in SMTMS properly as SSS 264. But on CRYO screens it is reported as SSS 269.
	ACR? ICS?
	
	

	Thursday_25_N
	TBD1
	SSS647:  Strength measurement and roll angle tests carried out at 2000A with SSW. The results were bad and were communicated to Marco (TC).
	MTM
	
	TC asked to send the result files by mail. Files have been sent. Tried to run roll angle thrice but the results were bad. Stages were misaligned by 6 mm wr MQY.
SSW adjusted Friday morning.

	Friday_26_M
	TBA1
	Display on TBA1 is showing the previous dipole name, MB2235 instead of MB1236.
	ACR
	
	Stephane contacted TC for action with J.Axensalva.

	Friday_26_M
	TBD1
	SSS647: SSW measurements of strength faulty quench outside the magnet (Ext-Vbus trigger).
	ACR
	
	A. Siemko asked cryo_team to increase the flow of cool down, because a problem of N line. Cryo team has been informed.

	Friday_26_N
	TBF1
	MB2279:Second training quench : alarm was reported in AQA (D1_L DirVT voltage exceeded 200V)
	MTM
	
	TC contacted. He advised to proceed with HV test and check with the warm _up quench.

OK, no alarm only warning.

	Friday_26_M
	TBA1
	MB1236: Extended test planned by Stephane after MM3. Launching the tests by TEMA are not working.
	MTM
	
	TC and Richard contacted. Congig files of the cycles must be modified with new versions of MMP.

Cleaning of the cycles will be performed (Action Guy, Stephane).

	Saturday_27_A
	TBE1
	MB2284: Problem in launching HV Test due to communication failure between PLCHV and PLCSE.
	AB/CO
	
	PLCSE needed RESET to launch CDBP.

	Saturday_27_N
	all
	Video display in control room stuck.
	
	
	

	Saturday_27_N
	TBD2
	SSS677: Communication problem between D.Bocian and OP team. Darius performed separately the warm_up quench. Change of shift occurred and details of the special test were not known.
	
	
	CQPT was not uploaded.

	Saturday_27_N
	TBB1
	SSS264: PT5.4 failed due to TCLAux_x1 signal problem. BP problem., oscillation.
	ACR?
	
	Test launched after 3 hours and many attempt.

	Sunday_28_M


	TBD1
	SSS647: SSW measurements, same problem as Friday. Quench in  Ext-Vbus after few minutes at high current.
	
	
	A. Siemko advised. TC will be contacted but according A. Siemko this magnet have specifique cable and it is not possible to make long measurement with current applied. Magnet put in stand_by up to Monday for investigation

	Monday_29_A
	All
	Video display is not working in Bellegarde side.
	
	
	




















